%%    TEMPLATE for articles submitted to the full-tex econf proceedings
%%     
%%
%%     Please do not remove lines commented out with %+
%%           these are for the editors' use.
%%
%%     Questions?  Send email to :  mpeskin@slac.stanford.edu
%%   

\documentclass[12pt]{article}        
\usepackage{epsfig}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%  Do not change these:
\textwidth=6.0in  \textheight=8.25in

%%  Adjust these for your printer:
\leftmargin=-0.3in   \topmargin=-0.20in


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  personal abbreviations and macros
%    the following package contains macros used in this document:
\input econfmacros.tex
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  To include an item in the INDEX of the conference volume,
%           flag it with    \index{<item name>}
%  The use of this macro is illustrated in the text.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\Title#1{\begin{center} {\Large {\bf #1} } \end{center}}

\begin{document}
\begin{thebibliography}{99}

%%
%%  bibliographic items can be constructed using the LaTeX format in SPIRES:
%%    see    http://www.slac.stanford.edu/spires/hep/latex.html
%%  SPIRES will also supply the CITATION line information; please include it.
%%

\bibitem{lhcb}
LHCb Collaboration, A. A. Alves Jr. {\emph et al.}, JINST {\bf 3}, S08005 (2008).

\bibitem{buras}
A. J. Buras {\emph et al.}, JHEP 1010 (2010).

\bibitem{bruyn}
K. de Bruyn {\emph et al.}, arXiV:1204.1737 (2012).

\bibitem{babu}
KS. Babu and C.F. Kolda, Phys. Rev. Lett. {\bf 84} 228 (2000).

\bibitem{bsmm}
LHCb Collaboration, Phys. Rev. Lett.{\bf 108} 231801 (2012).

\bibitem{burdman}
G. Burdman {\emph et al.}, Phys. Rev. D {\bf 66} 014009 (2002).

\bibitem{dmmbsm}
E. Golowich {\emph et al.}, Phys. Rev. D {\bf 79} (2009).

\bibitem{dmmlhcb}
LHCb Collaboration, LHCb-CONF-2012-005.

\bibitem{lhmtaummm}
M. Blanke {\emph et al.}, Acta Phys. Pol {\bf B41} 657 (2010).

\bibitem{taummmlhcb}
LHCb Collaboration, LHCb-CONF-2012-015.

\bibitem{angular}
C. Bobeth {\emph et al.}, JHEP {\bf 1107} 067 (2011).

\bibitem{angularlhcb}
LHCb Collaboration, LHCb-CONF-2012-008.

\bibitem{crossing}
M. Beneke {\emph et al.}, Eur. Phys. J {\bf C41} 173 (2005).\\
%A. Ali {\emph et al.}, Eur. Phys. J {\bf C47} 625 (2006).\\
C. Bobeth {\emph et al.}, JHEP {\bf 1201} 107 (2012).

\bibitem{isospin}
T. Feldman and J. Matias, JHEP {\bf 01} 074 (2003).

\bibitem{isospinlhcb}
LHCb Collaboration, JHEP {\bf 07} 133 (2012).

\bibitem{bpluspiplusmm}
S. Hai-Zhen {\emph et al.}, Com. Theor. Phys. {\bf 50} 696 (2008).

\bibitem{pipluslhcb}
LHCb Collaboration, LHCb-CONF-2012-006.

\bibitem{cpassym}
Y. Keum {\emph et al.}, Phys. Rev. D {\bf 72} 014013 (2005).

\bibitem{cpassymlhcb}
LHCb Collaboration, LHCb-CONF-2012-004.

\end{thebibliography}



\Title{Rare decays at LHCb}

\bigskip\bigskip

%+\addtocontents{toc}{{\it D. Reggiano}}
%+\label{ReggianoStart}

\begin{raggedright}  

COSME ADROVER PACHECO\index{Adrover, C.} on behalf of the LHCb Collaboration\\
{\it Centre de Physique des Particules de Marseille (CPPM)\\
Aix-Marseille Universit\'e, France}\\
\vskip 3mm
Rare leptonic and semileptonic decays are studied using 1.0\,fb$^{-1}$ of $pp$ collisions at a centre-of-mass energy of $\sqrt{s}=7$\,TeV collected by the LHCb experiment~\cite{lhcb} in 2011.
Branching fractions measurements, angular distributions and isospin asymmetries are presented using this data sample.
\bigskip\bigskip
\end{raggedright}



\section{Introduction}
      
Rare decays are excellent tests to infer the presence of physics beyond the Standard Model (BSM), as they occur through processes prohibited at
tree level in the SM. Any deviation from the SM prediction in branching fraction or angular distributions of such decays can lead to indications of new physics. 

\section{Branching fraction measurements of $B^{0}_{(s)} \to \mu^+ \mu^-$}
The SM predictions for the branching fractions of the decays \Bsmm\ and \Bdmm\ are small and have a low uncertainty:
$\mathcal{B}$(\Bsmm)$= (3.2 \pm 0.2)\times 10^{-9}$ and $\mathcal{B}$(\Bsmm)$= (0.10 \pm 0.01)\times 10^{-9}$~\cite{buras}.
Taking into account the oscillation of the $B^{0}_{s}$ system, the time integrated branching fraction is evaluated to be~\cite{bruyn}:
$\mathcal{B}$(\Bsmm)$= 3.4 \times 10^{-9}$.
However, these values can be significantly enhanced within Minimal Supersymmetric extensions of the SM (MSSM)~\cite{babu} due to contributions from
new processes or new heavy particles. The LHCb results are the most constraining limits~\cite{bsmm} on these branching fractions obtained with a single experiment to date at 95$\%$ C.L.: $\mathcal{B}(B^{0}_{(s)} \to \mu^+ \mu^-) < 4.5 \times 10^{-9}$ and $\mathcal{B}(B^{0} \to \mu^+ \mu^-) < 1.0 \times 10^{-10}$.   


\section{Branching fraction measurement of $D^{0} \to \mu^+ \mu^-$}
The decay \Dmm\ is very rare in the SM: $10^{-13}<\mathcal{B}(D^{0} \to \mu^+ \mu^-) < 6\times 10^{-11}$~\cite{burdman}. In the context of MSSM scenarios with R parity violation, the predicted branching fractions can be largely enhanced~\cite{dmmbsm}.
The LHCb measurement of the branching fraction of this decay is~\cite{dmmlhcb} $\mathcal{B}(D^{0} \to \mu^+ \mu^-) < 1.3\times 10^{-8}$ at 95$\%$ C.L., which is the world best limit to date.


\section{Branching fraction measurement of $\tau^- \to \mu^- \mu^+ \mu^-$}
The branching fraction of the decay \taummm\ is extremely suppressed in the SM.
In New Physics models this branching fraction can be substantially enhanced. For instance, in the context of Little Higgs models $\mathcal{B}(\tau^- \to \mu^- \mu^+ \mu^-)<10^{-7}$~\cite{lhmtaummm}.
The LHCb measurement of this branching fraction ($\mathcal{B}(\tau^- \to \mu^- \mu^+ \mu^-)<7.8 \times 10^{-8}$ at 95$\%$ C.L.)~\cite{taummmlhcb} is comparable with the current best limits. 


\section{Angular analysis of $B^{0} \to K^{*0} \mu^+ \mu^-$}
The angular distribution of the \BKsmm\ decay can be described with several $q^2$-dependent parameters that are predicted in the SM~\cite{angular} with high precision: A$_{\rm FB}$, F$_{\rm L}$, S$_{\rm 3}$, and A$_{\rm IM}$. The LHCb measurements of these quantities~\cite{angularlhcb} show no deviation from the SM predictions, and are the most precise measurements to date. Figure~\ref{fig:angular} shows the LHCb result for A$_{\rm FB}$ with the SM prediction superimposed. 

The region where the forward-backward asymmetry changes sign, known as zero-crossing point ($q^{2}_{0}$), is predicted to lie in the range [$4.0 < q^{2}_{0} < 4.3$\,GeV$^2$/c$^4$]~\cite{crossing}. LHCb has measured for the first time the zero-crossing point $q^{2}_{0} = 4.9^{+1.1}_{-1.3}$\,GeV$^2$/c$^4$, which agrees with the SM predictions (Figure~\ref{fig:angular}).


\begin{figure}[htb]
\begin{center}
\epsfig{file=afb_result,height=2in}
\epsfig{file=xpoint,height=2in}
\caption{Forward-backward asymmetry of the di-muon pair (left) and zero-crossing point (right).}
\label{fig:angular}
\end{center}
\end{figure}

%

\section{Isospin asymmetry of $B \to K^{(*)} \mu^+ \mu^-$}
The isospin asymmetry of the decays $B \to K \mu^+ \mu^-$ and $B \to K^{*} \mu^+ \mu^-$ is defined as:
\begin{equation}
A_I \equiv \frac{ \mathcal{B}(B^{0} \rightarrow K^{(*)0} \mu^+ \mu^-) - 
  \frac{\tau_0}{\tau_{\pm}}\mathcal{B}(B^{\pm} \rightarrow K^{(*)\pm} \mu^+ \mu^-)}
  { \mathcal{B}(B^{0} \rightarrow K^{(*)0} \mu^+ \mu^-) + 
  \frac{\tau_0}{\tau_{\pm}} \mathcal{B}(B^{\pm} \rightarrow K^{(*)\pm} \mu^+ \mu^-)},
\end{equation}
where $\frac{\tau_0}{\tau_{\pm}}$ is the ratio of lifetimes of the $B^{0}$ and $B^{\pm}$ mesons. 
For the $B \to K^{*} \mu^+ \mu^-$ case, the SM predicts $A_I$ of $\mathcal{O}(-1\%)$ at low $q^2$ and $\mathcal{O}(1\%)$ at $q^2 \to 0$~\cite{isospin},
while there is no precise prediction of $A_I$ for the $B \to K \mu^+ \mu^-$ case. 
Figure~\ref{fig:isospin} shows the LHCb results on the isospin asymmetries~\cite{isospinlhcb}. Theoretical prediction and experimental measurement are in agreement for the $B \to K^{*} \mu^+ \mu^-$ case, while a precise SM prediction for the $B \to K \mu^+ \mu^-$ case is foreseen in order to reveal whether the measured 4.4\,$\sigma$ statistical significance deviation from zero can be accommodated in the SM or is a New Physics effect.
 

\begin{figure}[htb]
\begin{center}
\epsfig{file=isospin_kstar_result,height=2in}
\epsfig{file=isospin_k_result,height=2in}
\caption{Isospin asymmetry as a function of $q^2$ in the $B \to K^{*} \mu^+ \mu^-$ (left) and $B \to K \mu^+ \mu^-$ systems (right).
The blue line corresponds to theoretical predictions.}
\label{fig:isospin}
\end{center}
\end{figure}
 
%

\section{Branching fraction measurement of $B^{+} \to \pi^{+} \mu^+ \mu^-$}
$B^{+} \to \pi^{+} \mu^+ \mu^-$ decays are processes mediated by $b \to d \,\,l^+ l^-$ transitions never observed before. 
In the SM, such transitions are suppressed by a factor $|V_{td}/V_{ts}|$ that does not necessarily appear in New Physics models.
The LHCb measurement~\cite{pipluslhcb}, $\mathcal{B}$($B^{+} \to \pi^{+} \mu^+ \mu^-$)$= (2.4 \pm 0.6_{(stat)} \pm 0.2_{(syst)})\times 10^{-8}$, is in agreement with the SM prediction of $\mathcal{B}$($B^{+} \to \pi^{+} \mu^+ \mu^-$)$= (1.91 \pm 0.21)\times 10^{-8}$~\cite{bpluspiplusmm}. $B^{+} \to \pi^{+} \mu^+ \mu^-$ is the rarest $B$ decay ever observed.


\section{Conclusions}
The LHCb Collaboration has presented measurements of branching fractions of beauty, charm, and tau decays, as well as of angular distribution of $B^{0} \to K^{*0} \mu^+ \mu^-$ and of the isospin asymmetry of the $B \to K^{(*)} \mu^+ \mu^-$ system.
All these measurements are of an unprecedented accuracy and are consistent with the SM predictions. 
The measured isospin asymmetry of the $B \to K \mu^+ \mu^-$ system shows a 4.4\,$\sigma$ deviation from zero.
 


%\def\Discussion{
%\setlength{\parskip}{0.3cm}\setlength{\parindent}{0.0cm}
%     \bigskip\bigskip      {\Large {\bf Discussion}} \bigskip}
%\def\speaker#1{{\bf #1:}\ }
%\def\endDiscussion{}

%\Discussion

%\speaker{D. Giovanni (University of Seville)}  My analysis indicates that the
%recovery of the two gentlemen is due simply to their embrace of the masculine
%principle and has nothing to do with magnetism at all.  Could you comment on 
%this?

%\speaker{Reggiano} Professor Giovanni has discussed this hypothesis in several
%forums, but, I do not believe there is anything in print.  I understand that
%he is spending his time in other pursuits.

%\speaker{D. Anna (University of Seville)}  In fact, my colleague Giovanni 
%has expressed opposite opinions on this question at various times, depending
%on the audience.  All of these testosterone-based theories are, of course,
%nonsense.

%\endDiscussion
 %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include an .eps figure into your paper
%%
%\begin{figure}[htb]
%\begin{center}
%\epsfig{file=magnet.eps,height=1.5in}
%\caption{Plan of the magnet used in the Mesmeric studies.}
%\label{fig:magnet}
%\end{center}
%\end{figure}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include a LaTeX table  into your paper
%%
%\begin{table}[b]
%\begin{center}
%\begin{tabular}{l|ccc}  
%Patient &  Initial level($\mu$g/cc) &  w. Magnet &  
%w. Magnet and Sound \\ \hline
% Guglielmo B.  &   0.12     &     0.10      &     0.001  \\
% Ferrando di N. &  0.15     &     0.11      &  $< 0.0005$ \\ \hline
%\end{tabular}
%\caption{Blood cyanide levels for the two patients.}
%\label{tab:blood}
%\end{center}
%\end{table}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%\bigskip
%I am grateful to Don Alfonso d'Alba for certain services essential to 
%this investigation.



\end{document}

